Author's response to reviews

Title: Increasing prevalence and high incidence of celiac disease in elderly people: A population-based study

Authors:

Anitta Vilppula (anitta.vilppula@phnet.fi)
Katri Kaukinen (katri.kaukinen@uta.fi)
Liisa Luostarinen (liisa.luostarinen@phsotey.fi)
Ilkka Krekelä (ilkka.krekela@phsotey.fi)
Heikki Patrikainen (heikki.patrikainen@phsotey.fi)
Raisa Valve (raisa.valve@helsinki.fi)
Markku Mäki (markku.maki@uta.fi)
Pekka Collin (pekka.collin@uta.fi)

Version: 3 Date: 15 June 2009

Author's response to reviews:

Major revisions:
Conclusions section has been inserted
Authors' contributions has been revised as suggested
Reference 4 has been completed
Figures have been submitted as Figure 1 and Figure 2; the references have been changed accordingly

Minor revisions:
email addresses have been included
Keywords have been removed
Patients and methods has been renamed as suggested
Coloured (highlighted) text has been formatted
Table legens have been revised